Author's response to reviews

Title: Splenic rupture as the presenting manifestation of primary splenic angiosarcoma in a teenage woman: a case report

Authors:

Andreas Manouras (amanouras@hippocratio.gr)
Panagiotis Giannopoulos (giannopoulosp@hotmail.com)
Levon Toufektzian (tlevon@gmail.com)
Haridimos Markogiannakis (markogiannakis@easy.com)
Emmanuel E Lagoudianakis (emlag@med.uoa.gr)
Artemisia Papadima (redemlag@yahoo.gr)
Dimitrios Papanikolaou (dpapanik81@yahoo.gr)
Konstantinos Filis (kfilis@hotmail.com)
Panagiotis Kekis (pkekis@hotmail.com)

Version: 6 Date: 21 April 2008

Author's response to reviews:

Dear Sir/Madam,

We would like to thank you for your decision and your help regarding our paper entitled “Splenic rupture as the presenting manifestation of primary splenic angiosarcoma in a teenage woman: a case report” (MS: 2274587151754306) which has been accepted for publication in the Journal of Medical Case Reports.

We have carefully checked the revised corrected proof of our article and believe it is excellent. We do confirm that we agree with all the amendments and English language corrections made and the manuscript can be published. In addition, we do confirm that all the sections highlighted in red are correct and we agree with them.

Please find attached the corrected proof of our article that you sent us.

There is only one minor correction to suggest:

In the “Discussion”, page 5, 2nd paragraph, the 4th sentence: “This mainly hematogenous tumor has a high incidence of early metastasis as reflected in the reported rates of between 69% and 100% among various series, with the most common sites being the liver, lungs, bones or bone marrow, lymph nodes, gastrointestinal tract, brain and adrenal glands.” should be changed to: “This tumor has a high incidence of early metastasis as reflected in the reported rates of between 69% and 100% among various series, with the most common sites being the liver, lungs, bones or bone marrow, lymph nodes, gastrointestinal tract, brain and adrenal glands.”.
We would like to thank you for your help regarding publication of this article. We
would be very grateful to you if you could please provide us with the final
corrected proof of our paper. We are looking forward to the publication of our
paper.

Yours sincerely,

Dr. Haridimos Markogiannakis, MD